Author’s response to reviews

Title: Scaling up community-based obesity prevention in Australia: Background and evaluation design of the Health Promoting Communities: Being Active Eating Well initiative.

Authors:

Andrea M de Silva-Sanigorski (andrea.sanigorski@deakin.edu.au)
Kristy Bolton (kabolton@deakin.edu.au)
Michelle Haby (michelle.haby@dhs.vic.gov.au)
Peter Kremer (peter.kremer@deakin.edu.au)
Lisa Gibbs (lgibbs@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Boyd Swinburn (boyd.swinburn@deakin.edu.au)

Version: 2 Date: 7 February 2010

Author’s response to reviews:

All revisions required have been made where possible. Two of the references remain 'In Press' and the publication details are not yet available.

regards
Andrea